Author's response to reviews

Title: The General Practitioner and the diagnosis of occupational diseases.

Authors:

   Luca Cegolon (mrcegolon@hotmail.com)
   John H Lange (jhlange1@hotmail.com)
   Giuseppe Mastrangelo (giuseppe.mastrangelo@unipd.it)

Version: 4 Date: 21 June 2010

Author's response to reviews:

Dear Editor,

I hereby re-submit the following paper entitled: "The Primary Care Practitioner and the Diagnosis of Occupational Diseases", by Cegolon L, et al.

The manuscript has been revised according to the requests of:

- Reviewer No.1 (amendments yellow marked);
- Editor (amendments violet marked);

Furthermore the language style of the body text has been improved (changes marked in blue), as recommended.

I hope this satisfactory, but please do not hesitate to contact me in case of further need.

I look forward to hearing from you soon.

Best Wishes.

Dr. Luca Cegolon, MD, FFPH

Corresponding Author